Author's response to reviews

Title: Phenobarbital versus Morphine in the management of Neonatal Abstinence Syndrome, a randomized control trial

Authors:

Fatemeh Nayeri (fnayeri@sina.tums.ac.ir)
Mahdi Sheikh (mahdisheikh@gmail.com)
Majid Kalani (mahduni@icloud.com)
Pedram Niknafs (Pniknaffs@yahoo.com)
Mamak Shariat (mshariat@tums.ac.ir)
Hosein Dalili (hoseindalili@yahoo.com)
Ahmad Reza Dehpour (dehpour@medscape.com)

Version: 2
Date: 31 July 2014

Author's response to reviews: see over
Dear Sir/Madam

Thank you very much for your kind consideration of our paper entitled "Phenobarbital versus Morphine in the management of Neonatal Abstinence Syndrome, a randomized control trial" with the manuscript number of "MS: 3802476441381809".

I received an email from the editorial office requesting adding page and line numbers to the manuscript, while the original uploaded manuscript had the page and line numbers. Here I am uploading the manuscript again with page numbers and line numbers.

Thank you again for kind consideration of our paper

Mahdi Sheikh, M.D.
Corresponding Author
Maternal, Fetal and Neonatal Research Center,
Tehran University of Medical Sciences,
Tehran, Iran